Author’s response to reviews

Title: Structured on-the-job training to improve retention of newborn resuscitation skills: a national cohort Helping Babies Breathe study in Tanzania

Authors:

Mary Drake (marylynndrake@yahoo.com)
Dunstan Bishanga (dunstan.bishanga@jhpiego.org)
Akwila Temu (akwilatemu@gmail.com)
Mustafa Njozi (mnjozi@gmail.com)
Erica Thomas (thomaserica@yahoo.com)
Victor Mponzi (vzmponzi@gmail.com)
Lauren Arlington (larlington@mgh.harvard.edu)
Georgina Msemo (georginamsemo@gmail.com)
Mary Azayo (mmazayo@gmail.com)
Allan Kairuki (allamwija@hotmail.com)
Amunga Meda (meda.robson@gmail.com)
Kahabi Isangula (kaisa079@yahoo.com)
Brett Nelson (brett.d.nelson@gmail.com)

Version: 3 Date: 15 Jan 2019

Author’s response to reviews:

Mary Drake
P.O. Box 9170
Dar es Salaam, Tanzania
Mary.Drake@Jhpiego.org
15 January 2019

BMC Pediatrics
Revision 3: Reviewer Comments & Author Responses
BPED-D-18-00605
Structured on-the-job training to improve retention of newborn resuscitation skills: a national cohort Helping Babies Breathe study in Tanzania
Mary Drake, BSN, MPH; Dunstan R. Bishanga; Akwila Temu; Mustafa Njozi; Erica Thomas; Victor Mponzi; Lauren Arlington; Georgina Msemo; Mary Azayo; Allan Kairuki; Amunga Meda; Kahabi G. Isangula; Brett D. Nelson

Dear Sezin Uzal and entire Reviewer Team,

On behalf of a team of co-authors from Tanzania and the United States, I am pleased to submit the updated version of the manuscript “Structured on-the-job training to improve retention of newborn resuscitation skills: a national cohort Helping Babies Breathe study in Tanzania”. We would like to thank the editor and reviewers for the positive feedback on the last version submitted, and for the support on the references. We have listed the reviewer comments and author responses below. I believe we have conformed to the journal requirements, however I am able to adjust if needed. We look forward to your positive feedback.

Comment 1:
Reference 20: The primary author is Ashish KC. The last name is KC, first name is Ashish. The formatting of his name is incorrect.

Comment 1 Author response:
A) Thank you for this observation. We have corrected the reference.

Comment 2
Another reference to consider is


Comment 2 Author response:
Thank you for this reference, we have added this to the Discussion section and into References (reference 29).

If the editor and reviewers have the possibility to review and share feedback in the month of January, the authors would be so grateful.

Kind regards,

Mary Drake